\begin{table}
\caption{Small Magellanic Cloud.}
%\centering
\par
\begin{tabular}{ccccl}
\hline\hline
\noalign{\smallskip}
Freq.     & Wavel.       & \multicolumn{2}{c}{Flux densities} \\
 $\nu$    & $\lambda$    &  Jy & $\Delta$Jy  & Reference\\
(GHz)     & (mm)         & \multicolumn{3}{c}{}\\
\noalign{\smallskip}
\hline
\noalign{\smallskip}
0.0197    & 15~230        &5270   & 1054 & Shain (\cite{Shain59}); cf. Alvarez et~al. (\cite{Alvarez89})\\
0.045     & 6670         & 415   &  80  & Alvarez et~al. (\cite{Alvarez89}) \\
0.0855    & 3510         & 460   & 200  & Mills (1959)\tablefootmark{a} \\
0.408     & 735          & 133   &  10  & Haslam (1982)\tablefootmark{a}  \\
0.843     & 356          & 75    &  8   & Ye \& Turtle (\cite{Ye91}) \\
1.4 	  & 214          & 42    &  6   & 
Loiseau et~al. (\cite{Loiseau87})\tablefootmark{b}  \\
2.3       & 130          & 31    &  6   & Mountfort et~al. (\cite{Mountfort87}) \\
2.45      & 122          & 26    &  3   & Haynes et al. (\cite{Haynes91}) \\
4.75      & 63           & 19    &  4   & Haynes et al. (\cite{Haynes91}) \\
8.55      & 35           & 15    &  4   & Haynes et al. (\cite{Haynes91}) \\
23	  & 13           & 17    &  3   & This paper (2010) \\
33	  & 9.1          & 20    &  4   & This paper (2010) \\
41	  & 7.3          & 24    &  5   & This paper (2010) \\
61	  & 4.9          & 40    &  8   & This paper (2010) \\
93	  & 3.2          & 88    &  18  & This paper (2010) \\         
245	  & 1.2          & 320   &  80  & Aguirre et~al. (\cite{Aguirre03}) \\
400	  & 0.75         & 950   & 190  & Aguirre et~al. (\cite{Aguirre03}) \\
460	  & 0.65         & 1620  & 290  & Aguirre et~al. (\cite{Aguirre03}) \\
630	  & 0.48         & 3200  & 810  & Aguirre et~al. (\cite{Aguirre03}) \\
1250	  & 0.24         & 12~070 & 540  & Aguirre et~al. (\cite{Aguirre03}) \\
1250	  & 0.24         & 12~350 & 1440 & This paper (2010) \\
1250	  & 0.24         & 9600  & 4400 & Stanimirovic et~al. (\cite{Stanimirovic00}) \\
1765      & 0.17  1      & 18~500 & 4000 & Wilke et al. (\cite{Wilke04})\tablefootmark{c}  \\
1875	  & 0.16         & 19~000 & 4300 & Leroy et~al. (\cite{Leroy07}) \\
2140	  & 0.14         & 20~300 & 920  & Aguirre et~al. (\cite{Aguirre03}) \\
2140	  & 0.14         & 18~900 & 2005 & This paper (2010) \\
2140	  & 0.14         & 14~000 & 5600 & Stanimirovic et~al. (\cite{Stanimirovic00}) \\
3000	  & 0.10         & 16~480 & 2220 & Aguirre et~al. (\cite{Aguirre03}) \\
3000	  & 0.10         & 15~750 & 2125 & This paper (2010) \\
3000	  & 0.10         & 13~600 & 930  & Stanimirovic et~al. (\cite{Stanimirovic00}) \\
3000	  & 0.10         & 13~900 & 1810 & Schwering (\cite{Schwering88}) \\
4285	  & 0.07         & 12~500 & 2400 & Leroy et~al. (\cite{Leroy07}) \\
5000	  & 0.06         & 8450  & 730  & This paper \\
5000	  & 0.06         & 6700  & 1060 & Stanimirovic et~al. (\cite{Stanimirovic00}) \\
5000	  & 0.06         & 7170  & 957  & Schwering (\cite{Schwering88}) \\
12~500	  & 0.024        & 350   & 50   & Leroy et~al. (\cite{Leroy07}) \\
12~000	  & 0.025        & 460   & 180  & Stanimirovic et~al. (\cite{Stanimirovic00}) \\
12~000	  & 0.025        & 425   & 70   & This paper \\
12~000	  & 0.025        & 310   & 170  & Stanimirovic et~al. (\cite{Stanimirovic00}) \\
12~000	  & 0.025        & 385   & 56   & Schwering (\cite{Schwering88}) \\
25~000	  & 0.012        & 80    & 30   & Stanimirovic et~al. (\cite{Stanimirovic00}) \\
25~000	  & 0.012        & 105   & 30   & Stanimirovic et~al. (\cite{Stanimirovic00}) \\
25~000	  & 0.012        & 147   & 26   & Stanimirovic et~al. (\cite{Stanimirovic00}) \\
25~000     & 0.012        & 235   & 25   & This paper (2010) \\
37~500	  & 0.008        & 78    & 8    & Bolatto et~al. (\cite{Bolatto07}) \\
51~274	  & 0.0058       & 85    & 9    & Bolatto et~al. (\cite{Bolatto07}) \\
61~220	  & 0.0049       & 110   & 43   & Stanimirovic et~al. (\cite{Stanimirovic00}) \\
61~220     & 0.0049       & 150   & 25   & This paper (2010) \\
66~667	  & 0.0045       & 115   & 12   & Bolatto et~al. (\cite{Bolatto07}) \\
83~333	  & 0.0036       & 163   & 16   & Bolatto et~al. (\cite{Bolatto07}) \\
85~710	  & 0.0035       & 220   & 31   & Stanimirovic et~al. (\cite{Stanimirovic00}) \\
85~710	  & 0.0035       & 280   & 40   & This paper (2010) \\
136~365	  & 0.0022       & 400   & 37   & Stanimirovic et~al. (\cite{Stanimirovic00}) \\
136~365	  & 0.0022       & 525   & 65   & This paper (2010) \\
240~000	  & 0.00125      & 510   & 25   & Stanimirovic et~al. (\cite{Stanimirovic00}) \\
240~000	  & 0.00125      & 670   & 80   & This paper (2010) \\
542~000	  & 0.00055      & 458   & 46   & de Vaucouleurs et~al. 
(\cite{deVaucouleurs91}) \\
681~000	  & 0.00044      & 354   & 34   & de Vaucouleurs et~al. 
(\cite{deVaucouleurs91}) \\
819~000    & 0.000365     & 181   & 22   & de Vaucouleurs et~al. (\cite{deVaucouleurs91}) \\
\noalign{\smallskip}
\hline
\noalign{\smallskip}
\end{tabular}
\tablefoot {\tablefoottext{a}{As revised and listed by Loiseau et~al. (\cite{Loiseau87})}; 
\tablefoottext{b}{As revised and listed by Haynes et~al. (\cite{Haynes91});}
\tablefoottext{c}{As revised by Leroy et~al. (\cite{Leroy07}).}}
\end{table}